Author's response to reviews

Title: Impact of the introduction of pneumococcal conjugate vaccine on immunization coverage among infants

Authors:
- Nancy D Lin (nlin@stanford.edu)
- Ken Kleinman (ken_kleinman@harvardpilgrim.org)
- K ARNOLD Chan (kachan@hsph.harvard.edu)
- Xian-Jie Yu (yuuxi2002@yahoo.com)
- Eric K France (eric.k.france@kp.org)
- Stanley Xu (stan.xu@kp.org)
- Feifei Wei (feifei.wei@healthpartners.com)
- John Mullooly (john.mullooly@kpchr.org)
- Jeanne Santoli (zmd4@cdc.gov)
- Tracy A Lieu (tracy_lieu@harvardpilgrim.org)

Version: 5 Date: 18 November 2005

Author's response to reviews: see over
Center for Primary Care and Outcomes Research  
117 Encina Commons  
Stanford, CA 94305-6019

November 18, 2005

BioMed Central Pediatrics  
Middlesex House  
32-42 Cleveland Street  
London  
W1T 4LB  
UK

Dear Editor:

As requested, we have made the following formatting changes to the manuscript, “Impact of the introduction of pneumococcal conjugate vaccine on immunization coverage among infants” (MS 1283885339736012):

- References – In the manuscript text, references have been placed before the full stop.
- References - The reference list includes all authors’ names, in place of the term et al.
- EndNote references – Any edits to references were made in the Endnote record prior to creation of the reference list.
- Tables – Tables were modified to remove merging of cells across rows.

Thank you again for your review. If any further changes are required, please do not hesitate to contact me.

Sincerely,

Nancy Lin, M.Sc., Sc.D.